---
title: 'Overview'
description: 'Object schemas extracted from the OpenAPI schema.'
---

This section contains all resource schemas (data types) used by the Fluxer API.

## Field notation

Resource tables use a compact notation:

| Notation | Meaning |
|----------|----------|
| `field` | Required field |
| `field?` | Optional field (may be omitted) |
| `?type` | Nullable (value can be `null`) |
| `field?` with `?type` | Optional and nullable |

## Resources by domain

- [Admin](/resources/admin) (169 schemas)
- [Auth](/resources/auth) (37 schemas)
- [Billing](/resources/billing) (3 schemas)
- [Channels](/resources/channels) (66 schemas)
- [Common](/resources/common) (35 schemas)
- [Gateway](/resources/gateway) (2 schemas)
- [Gifts](/resources/gifts) (1 schemas)
- [Guilds](/resources/guilds) (63 schemas)
- [Instance](/resources/instance) (1 schemas)
- [Invites](/resources/invites) (4 schemas)
- [Klipy](/resources/klipy) (5 schemas)
- [Oauth2](/resources/oauth2) (25 schemas)
- [Packs](/resources/packs) (7 schemas)
- [Premium](/resources/premium) (2 schemas)
- [Read states](/resources/read_states) (1 schemas)
- [Reports](/resources/reports) (14 schemas)
- [Saved media](/resources/saved_media) (5 schemas)
- [Search](/resources/search) (2 schemas)
- [Themes](/resources/themes) (2 schemas)
- [Users](/resources/users) (81 schemas)
- [Webhooks](/resources/webhooks) (9 schemas)

## All schemas

- [AddSnowflakeReservationRequest](/resources/admin#addsnowflakereservationrequest)
- [AdminArchiveResponseSchema](/resources/admin#adminarchiveresponseschema)
- [AdminAuditLogResponseSchema](/resources/common#adminauditlogresponseschema)
- [AdminUserDmChannelSchema](/resources/common#adminuserdmchannelschema)
- [AdminUsersMeResponse](/resources/admin#adminusersmeresponse)
- [APIErrorCode](/resources/common#apierrorcode)
- [ApplicationBotResponse](/resources/common#applicationbotresponse)
- [ApplicationCreateRequest](/resources/oauth2#applicationcreaterequest)
- [ApplicationFlags](/resources/common#applicationflags)
- [ApplicationListResponse](/resources/oauth2#applicationlistresponse)
- [ApplicationPublicResponse](/resources/oauth2#applicationpublicresponse)
- [ApplicationResponse](/resources/oauth2#applicationresponse)
- [ApplicationsMeResponse](/resources/oauth2#applicationsmeresponse)
- [ApplicationUpdateRequest](/resources/oauth2#applicationupdaterequest)
- [AuditLogActionType](/resources/guilds#auditlogactiontype)
- [AuditLogChangeSchema](/resources/common#auditlogchangeschema)
- [AuditLogsListResponseSchema](/resources/admin#auditlogslistresponseschema)
- [AuditLogWebhookResponse](/resources/common#auditlogwebhookresponse)
- [AuthenticatorType](/resources/common#authenticatortype)
- [AuthLoginResponse](/resources/auth#authloginresponse)
- [AuthMfaRequiredResponse](/resources/common#authmfarequiredresponse)
- [AuthorizeConsentRequest](/resources/oauth2#authorizeconsentrequest)
- [AuthorizeIpRequest](/resources/auth#authorizeiprequest)
- [AuthRegisterResponse](/resources/auth#authregisterresponse)
- [AuthSessionClientInfo](/resources/common#authsessionclientinfo)
- [AuthSessionLocation](/resources/common#authsessionlocation)
- [AuthSessionResponse](/resources/common#authsessionresponse)
- [AuthSessionsResponse](/resources/auth#authsessionsresponse)
- [AuthTokenWithUserIdResponse](/resources/auth#authtokenwithuseridresponse)
- [BanCheckResponseSchema](/resources/admin#bancheckresponseschema)
- [BanEmailRequest](/resources/admin#banemailrequest)
- [BanGuildMemberRequest](/resources/admin#banguildmemberrequest)
- [BanIpRequest](/resources/admin#baniprequest)
- [BanPhoneRequest](/resources/admin#banphonerequest)
- [Base64ImageType](/resources/common#base64imagetype)
- [BlueskyAuthorizeRequest](/resources/common#blueskyauthorizerequest)
- [BlueskyAuthorizeResponse](/resources/common#blueskyauthorizeresponse)
- [BotFlags](/resources/common#botflags)
- [BotProfileResponse](/resources/oauth2#botprofileresponse)
- [BotProfileUpdateRequest](/resources/oauth2#botprofileupdaterequest)
- [BotTokenResetResponse](/resources/oauth2#bottokenresetresponse)
- [BulkAddGuildMembersRequest](/resources/admin#bulkaddguildmembersrequest)
- [BulkDeleteMessagesRequest](/resources/channels#bulkdeletemessagesrequest)
- [BulkOperationFailedResponse](/resources/common#bulkoperationfailedresponse)
- [BulkOperationResponse](/resources/admin#bulkoperationresponse)
- [BulkScheduleUserDeletionRequest](/resources/admin#bulkscheduleuserdeletionrequest)
- [BulkUpdateGuildFeaturesRequest](/resources/admin#bulkupdateguildfeaturesrequest)
- [BulkUpdateUserFlagsRequest](/resources/admin#bulkupdateuserflagsrequest)
- [CallEligibilityResponse](/resources/channels#calleligibilityresponse)
- [CallRingBodySchema](/resources/channels#callringbodyschema)
- [CallUpdateBodySchema](/resources/channels#callupdatebodyschema)
- [CancelBulkMessageDeletionRequest](/resources/admin#cancelbulkmessagedeletionrequest)
- [ChangeDobRequest](/resources/admin#changedobrequest)
- [ChangeEmailRequest](/resources/admin#changeemailrequest)
- [ChangeUsernameRequest](/resources/admin#changeusernamerequest)
- [ChannelCreateCategoryRequest](/resources/common#channelcreatecategoryrequest)
- [ChannelCreateLinkRequest](/resources/common#channelcreatelinkrequest)
- [ChannelCreateRequest](/resources/guilds#channelcreaterequest)
- [ChannelCreateTextRequest](/resources/common#channelcreatetextrequest)
- [ChannelCreateVoiceRequest](/resources/common#channelcreatevoicerequest)
- [ChannelInviteCreateRequest](/resources/invites#channelinvitecreaterequest)
- [ChannelNicknameOverrides](/resources/common#channelnicknameoverrides)
- [ChannelOverwriteRequest](/resources/common#channeloverwriterequest)
- [ChannelOverwriteResponse](/resources/common#channeloverwriteresponse)
- [ChannelPartialRecipientResponse](/resources/common#channelpartialrecipientresponse)
- [ChannelPartialResponse](/resources/common#channelpartialresponse)
- [ChannelPinMessageResponse](/resources/common#channelpinmessageresponse)
- [ChannelPinResponse](/resources/common#channelpinresponse)
- [ChannelPinsResponse](/resources/channels#channelpinsresponse)
- [ChannelPositionUpdateRequest](/resources/guilds#channelpositionupdaterequest)
- [ChannelResponse](/resources/channels#channelresponse)
- [ChannelUpdateCategoryRequest](/resources/common#channelupdatecategoryrequest)
- [ChannelUpdateGroupDmRequest](/resources/common#channelupdategroupdmrequest)
- [ChannelUpdateLinkRequest](/resources/common#channelupdatelinkrequest)
- [ChannelUpdateRequest](/resources/channels#channelupdaterequest)
- [ChannelUpdateTextRequest](/resources/common#channelupdatetextrequest)
- [ChannelUpdateVoiceRequest](/resources/common#channelupdatevoicerequest)
- [ClearGuildFieldsRequest](/resources/admin#clearguildfieldsrequest)
- [ClearUserFieldsRequest](/resources/admin#clearuserfieldsrequest)
- [CodesResponse](/resources/admin#codesresponse)
- [ConnectionListResponse](/resources/common#connectionlistresponse)
- [ConnectionResponse](/resources/common#connectionresponse)
- [ConnectionVerificationResponse](/resources/common#connectionverificationresponse)
- [CreateAdminApiKeyRequest](/resources/admin#createadminapikeyrequest)
- [CreateAdminApiKeyResponse](/resources/admin#createadminapikeyresponse)
- [CreateCheckoutSessionRequest](/resources/billing#createcheckoutsessionrequest)
- [CreateConnectionRequest](/resources/common#createconnectionrequest)
- [CreateFavoriteMemeBodySchema](/resources/saved_media#createfavoritememebodyschema)
- [CreateFavoriteMemeFromUrlBodySchema](/resources/saved_media#createfavoritememefromurlbodyschema)
- [CreatePrivateChannelRequest](/resources/users#createprivatechannelrequest)
- [CreateSystemDmJobRequest](/resources/admin#createsystemdmjobrequest)
- [CreateVoiceRegionRequest](/resources/admin#createvoiceregionrequest)
- [CreateVoiceRegionResponse](/resources/admin#createvoiceregionresponse)
- [CreateVoiceServerRequest](/resources/admin#createvoiceserverrequest)
- [CreateVoiceServerResponse](/resources/admin#createvoiceserverresponse)
- [CustomStatusPayload](/resources/common#customstatuspayload)
- [CustomStatusResponse](/resources/common#customstatusresponse)
- [DefaultMessageNotifications](/resources/common#defaultmessagenotifications)
- [DeleteAllUserMessagesRequest](/resources/admin#deleteallusermessagesrequest)
- [DeleteAllUserMessagesResponse](/resources/admin#deleteallusermessagesresponse)
- [DeleteApiKeyResponse](/resources/admin#deleteapikeyresponse)
- [DeleteGuildRequest](/resources/admin#deleteguildrequest)
- [DeleteMessageRequest](/resources/admin#deletemessagerequest)
- [DeleteMessageResponse](/resources/admin#deletemessageresponse)
- [DeleteSnowflakeReservationRequest](/resources/admin#deletesnowflakereservationrequest)
- [DeleteVoiceRegionRequest](/resources/admin#deletevoiceregionrequest)
- [DeleteVoiceResponse](/resources/admin#deletevoiceresponse)
- [DeleteVoiceServerRequest](/resources/admin#deletevoiceserverrequest)
- [DisableForSuspiciousActivityRequest](/resources/admin#disableforsuspiciousactivityrequest)
- [DisableMfaRequest](/resources/admin#disablemfarequest)
- [DisableTotpRequest](/resources/users#disabletotprequest)
- [DiscoveryAdminRejectRequest](/resources/admin#discoveryadminrejectrequest)
- [DiscoveryAdminRemoveRequest](/resources/admin#discoveryadminremoverequest)
- [DiscoveryAdminReviewRequest](/resources/admin#discoveryadminreviewrequest)
- [DiscoveryApplicationPatchRequest](/resources/common#discoveryapplicationpatchrequest)
- [DiscoveryApplicationRequest](/resources/common#discoveryapplicationrequest)
- [DiscoveryApplicationResponse](/resources/admin#discoveryapplicationresponse)
- [DiscoveryCategoryListResponse](/resources/common#discoverycategorylistresponse)
- [DiscoveryCategoryResponse](/resources/common#discoverycategoryresponse)
- [DiscoveryGuildListResponse](/resources/common#discoveryguildlistresponse)
- [DiscoveryGuildResponse](/resources/common#discoveryguildresponse)
- [DiscoveryStatusResponse](/resources/common#discoverystatusresponse)
- [DonationCheckoutRequest](/resources/common#donationcheckoutrequest)
- [DonationCheckoutResponse](/resources/common#donationcheckoutresponse)
- [DonationRequestLinkRequest](/resources/common#donationrequestlinkrequest)
- [DownloadUrlResponseSchema](/resources/admin#downloadurlresponseschema)
- [DsaReportEmailSendRequest](/resources/reports#dsareportemailsendrequest)
- [DsaReportEmailVerifyRequest](/resources/reports#dsareportemailverifyrequest)
- [DsaReportGuildRequest](/resources/common#dsareportguildrequest)
- [DsaReportMessageRequest](/resources/common#dsareportmessagerequest)
- [DsaReportRequest](/resources/reports#dsareportrequest)
- [DsaReportUserRequest](/resources/common#dsareportuserrequest)
- [EmailChangeBouncedRequestNewRequest](/resources/users#emailchangebouncedrequestnewrequest)
- [EmailChangeBouncedVerifyNewRequest](/resources/users#emailchangebouncedverifynewrequest)
- [EmailChangeRequestNewRequest](/resources/users#emailchangerequestnewrequest)
- [EmailChangeRequestNewResponse](/resources/users#emailchangerequestnewresponse)
- [EmailChangeStartResponse](/resources/users#emailchangestartresponse)
- [EmailChangeTicketRequest](/resources/users#emailchangeticketrequest)
- [EmailChangeVerifyNewRequest](/resources/users#emailchangeverifynewrequest)
- [EmailChangeVerifyOriginalRequest](/resources/users#emailchangeverifyoriginalrequest)
- [EmailChangeVerifyOriginalResponse](/resources/users#emailchangeverifyoriginalresponse)
- [EmailRevertRequest](/resources/auth#emailrevertrequest)
- [EmailTokenResponse](/resources/users#emailtokenresponse)
- [EmailType](/resources/common#emailtype)
- [EmbedAuthorResponse](/resources/common#embedauthorresponse)
- [EmbedFieldResponse](/resources/common#embedfieldresponse)
- [EmbedFooterResponse](/resources/common#embedfooterresponse)
- [EmbedMediaFlags](/resources/common#embedmediaflags)
- [EmbedMediaResponse](/resources/common#embedmediaresponse)
- [EmptyBodyRequest](/resources/users#emptybodyrequest)
- [EnabledToggleRequest](/resources/guilds#enabledtogglerequest)
- [EnableMfaTotpRequest](/resources/users#enablemfatotprequest)
- [Error](/resources/admin#error)
- [ExpandVisionarySlotsRequest](/resources/admin#expandvisionaryslotsrequest)
- [FavoriteMemeListResponse](/resources/saved_media#favoritememelistresponse)
- [FavoriteMemeResponse](/resources/saved_media#favoritememeresponse)
- [ForceAddUserToGuildRequest](/resources/admin#forceaddusertoguildrequest)
- [ForgotPasswordRequest](/resources/auth#forgotpasswordrequest)
- [FriendRequestByTagRequest](/resources/users#friendrequestbytagrequest)
- [FriendSourceFlags](/resources/common#friendsourceflags)
- [GatewayBotResponse](/resources/gateway#gatewaybotresponse)
- [GenerateGiftCodesRequest](/resources/admin#generategiftcodesrequest)
- [GetArchiveResponseSchema](/resources/admin#getarchiveresponseschema)
- [GetIndexRefreshStatusRequest](/resources/admin#getindexrefreshstatusrequest)
- [GetProcessMemoryStatsRequest](/resources/admin#getprocessmemorystatsrequest)
- [GetVoiceRegionRequest](/resources/admin#getvoiceregionrequest)
- [GetVoiceRegionResponse](/resources/admin#getvoiceregionresponse)
- [GetVoiceServerRequest](/resources/admin#getvoiceserverrequest)
- [GetVoiceServerResponse](/resources/admin#getvoiceserverresponse)
- [GiftCodeMetadataResponse](/resources/users#giftcodemetadataresponse)
- [GiftCodeResponse](/resources/gifts#giftcoderesponse)
- [GitHubWebhook](/resources/webhooks#githubwebhook)
- [GlobalSearchMessagesRequest](/resources/search#globalsearchmessagesrequest)
- [GroupDmAddPermissionFlags](/resources/common#groupdmaddpermissionflags)
- [GroupDmInviteMetadataResponse](/resources/common#groupdminvitemetadataresponse)
- [GroupDmInviteResponse](/resources/common#groupdminviteresponse)
- [GuildAdminResponse](/resources/common#guildadminresponse)
- [GuildAssetItemSchema](/resources/common#guildassetitemschema)
- [GuildAuditLogEntryResponse](/resources/common#guildauditlogentryresponse)
- [GuildAuditLogListResponse](/resources/guilds#guildauditloglistresponse)
- [GuildBanCreateRequest](/resources/guilds#guildbancreaterequest)
- [GuildBanResponse](/resources/guilds#guildbanresponse)
- [GuildCreateRequest](/resources/guilds#guildcreaterequest)
- [GuildDeleteRequest](/resources/guilds#guilddeleterequest)
- [GuildEmojiBulkCreateRequest](/resources/guilds#guildemojibulkcreaterequest)
- [GuildEmojiBulkCreateResponse](/resources/guilds#guildemojibulkcreateresponse)
- [GuildEmojiCreateRequest](/resources/guilds#guildemojicreaterequest)
- [GuildEmojiResponse](/resources/guilds#guildemojiresponse)
- [GuildEmojiUpdateRequest](/resources/guilds#guildemojiupdaterequest)
- [GuildEmojiWithUserListResponse](/resources/guilds#guildemojiwithuserlistresponse)
- [GuildEmojiWithUserResponse](/resources/common#guildemojiwithuserresponse)
- [GuildExplicitContentFilter](/resources/common#guildexplicitcontentfilter)
- [GuildFeatureSchema](/resources/common#guildfeatureschema)
- [GuildFolderFlags](/resources/common#guildfolderflags)
- [GuildInviteMetadataResponse](/resources/common#guildinvitemetadataresponse)
- [GuildInviteResponse](/resources/common#guildinviteresponse)
- [GuildMemberProfileFlags](/resources/common#guildmemberprofileflags)
- [GuildMemberResponse](/resources/guilds#guildmemberresponse)
- [GuildMemberSearchRequest](/resources/guilds#guildmembersearchrequest)
- [GuildMemberSearchResponse](/resources/guilds#guildmembersearchresponse)
- [GuildMemberSearchResult](/resources/common#guildmembersearchresult)
- [GuildMemberUpdateRequest](/resources/guilds#guildmemberupdaterequest)
- [GuildMemoryStatsResponse](/resources/admin#guildmemorystatsresponse)
- [GuildMFALevel](/resources/common#guildmfalevel)
- [GuildOperations](/resources/common#guildoperations)
- [GuildReportCategoryEnum](/resources/common#guildreportcategoryenum)
- [GuildResponse](/resources/guilds#guildresponse)
- [GuildRoleCreateRequest](/resources/guilds#guildrolecreaterequest)
- [GuildRoleHoistPositionItem](/resources/common#guildrolehoistpositionitem)
- [GuildRoleHoistPositionsRequest](/resources/guilds#guildrolehoistpositionsrequest)
- [GuildRolePositionItem](/resources/common#guildrolepositionitem)
- [GuildRolePositionsRequest](/resources/guilds#guildrolepositionsrequest)
- [GuildRoleResponse](/resources/guilds#guildroleresponse)
- [GuildRoleUpdateRequest](/resources/guilds#guildroleupdaterequest)
- [GuildStickerBulkCreateRequest](/resources/guilds#guildstickerbulkcreaterequest)
- [GuildStickerBulkCreateResponse](/resources/guilds#guildstickerbulkcreateresponse)
- [GuildStickerCreateRequest](/resources/guilds#guildstickercreaterequest)
- [GuildStickerResponse](/resources/guilds#guildstickerresponse)
- [GuildStickerUpdateRequest](/resources/guilds#guildstickerupdaterequest)
- [GuildStickerWithUserListResponse](/resources/guilds#guildstickerwithuserlistresponse)
- [GuildStickerWithUserResponse](/resources/common#guildstickerwithuserresponse)
- [GuildTransferOwnershipRequest](/resources/guilds#guildtransferownershiprequest)
- [GuildUpdateRequest](/resources/guilds#guildupdaterequest)
- [GuildUpdateResponse](/resources/admin#guildupdateresponse)
- [GuildVanityURLResponse](/resources/guilds#guildvanityurlresponse)
- [GuildVanityURLUpdateRequest](/resources/guilds#guildvanityurlupdaterequest)
- [GuildVanityURLUpdateResponse](/resources/guilds#guildvanityurlupdateresponse)
- [GuildVerificationLevel](/resources/common#guildverificationlevel)
- [HandoffCompleteRequest](/resources/auth#handoffcompleterequest)
- [HandoffInitiateResponse](/resources/auth#handoffinitiateresponse)
- [HandoffStatusResponse](/resources/auth#handoffstatusresponse)
- [HarvestCreationResponseSchema](/resources/users#harvestcreationresponseschema)
- [HarvestDownloadUrlResponse](/resources/users#harvestdownloadurlresponse)
- [HarvestStatusEnum](/resources/common#harveststatusenum)
- [HarvestStatusResponseSchema](/resources/users#harveststatusresponseschema)
- [HarvestStatusResponseSchemaNullable](/resources/users#harveststatusresponseschemanullable)
- [IncomingCallFlags](/resources/common#incomingcallflags)
- [IndexRefreshStatusResponse](/resources/admin#indexrefreshstatusresponse)
- [InstanceConfigResponse](/resources/admin#instanceconfigresponse)
- [InstanceConfigUpdateRequest](/resources/admin#instanceconfigupdaterequest)
- [Int32Type](/resources/guilds#int32type)
- [Int64StringType](/resources/common#int64stringtype)
- [Int64Type](/resources/common#int64type)
- [IntrospectRequestForm](/resources/oauth2#introspectrequestform)
- [InviteMetadataResponseSchema](/resources/invites#invitemetadataresponseschema)
- [InviteResponseSchema](/resources/invites#inviteresponseschema)
- [IpAuthorizationPollResponse](/resources/auth#ipauthorizationpollresponse)
- [IpBanListEntrySchema](/resources/common#ipbanlistentryschema)
- [JoinSourceType](/resources/common#joinsourcetype)
- [KickGuildMemberRequest](/resources/admin#kickguildmemberrequest)
- [KlipyCategoryTagResponse](/resources/common#klipycategorytagresponse)
- [KlipyFeaturedResponse](/resources/klipy#klipyfeaturedresponse)
- [KlipyGifResponse](/resources/klipy#klipygifresponse)
- [KlipyRegisterShareRequest](/resources/klipy#klipyregistersharerequest)
- [LegalHoldRequest](/resources/admin#legalholdrequest)
- [LegalHoldResponse](/resources/admin#legalholdresponse)
- [LimitConfigGetResponse](/resources/admin#limitconfiggetresponse)
- [LimitConfigUpdateRequest](/resources/admin#limitconfigupdaterequest)
- [LimitFilterResponse](/resources/common#limitfilterresponse)
- [LimitRuleResponse](/resources/common#limitruleresponse)
- [ListAdminApiKeyResponse](/resources/admin#listadminapikeyresponse)
- [ListArchivesRequest](/resources/admin#listarchivesrequest)
- [ListArchivesResponseSchema](/resources/admin#listarchivesresponseschema)
- [ListAuditLogsRequest](/resources/admin#listauditlogsrequest)
- [ListBansRequest](/resources/admin#listbansrequest)
- [ListEmailBansResponseSchema](/resources/admin#listemailbansresponseschema)
- [ListGuildEmojisResponse](/resources/admin#listguildemojisresponse)
- [ListGuildMembersRequest](/resources/admin#listguildmembersrequest)
- [ListGuildMembersResponse](/resources/admin#listguildmembersresponse)
- [ListGuildStickersResponse](/resources/admin#listguildstickersresponse)
- [ListIpBansResponseSchema](/resources/admin#listipbansresponseschema)
- [ListPhoneBansResponseSchema](/resources/admin#listphonebansresponseschema)
- [ListReportsRequest](/resources/admin#listreportsrequest)
- [ListReportsResponse](/resources/admin#listreportsresponse)
- [ListSnowflakeReservationsResponse](/resources/admin#listsnowflakereservationsresponse)
- [ListSystemDmJobsResponse](/resources/admin#listsystemdmjobsresponse)
- [ListUserChangeLogRequest](/resources/admin#listuserchangelogrequest)
- [ListUserChangeLogResponseSchema](/resources/admin#listuserchangelogresponseschema)
- [ListUserDmChannelsRequest](/resources/admin#listuserdmchannelsrequest)
- [ListUserDmChannelsResponse](/resources/admin#listuserdmchannelsresponse)
- [ListUserGuildsRequest](/resources/admin#listuserguildsrequest)
- [ListUserGuildsResponse](/resources/admin#listuserguildsresponse)
- [ListUserSessionsRequest](/resources/admin#listusersessionsrequest)
- [ListUserSessionsResponse](/resources/admin#listusersessionsresponse)
- [ListVisionarySlotsResponse](/resources/admin#listvisionaryslotsresponse)
- [ListVoiceRegionsRequest](/resources/admin#listvoiceregionsrequest)
- [ListVoiceRegionsResponse](/resources/admin#listvoiceregionsresponse)
- [ListVoiceServersRequest](/resources/admin#listvoiceserversrequest)
- [ListVoiceServersResponse](/resources/admin#listvoiceserversresponse)
- [Locale](/resources/klipy#locale)
- [LoginRequest](/resources/auth#loginrequest)
- [LogoutAuthSessionsRequest](/resources/auth#logoutauthsessionsrequest)
- [LookupGuildRequest](/resources/admin#lookupguildrequest)
- [LookupGuildResponse](/resources/admin#lookupguildresponse)
- [LookupMessageByAttachmentRequest](/resources/admin#lookupmessagebyattachmentrequest)
- [LookupMessageRequest](/resources/admin#lookupmessagerequest)
- [LookupMessageResponse](/resources/admin#lookupmessageresponse)
- [LookupUserByIdsRequest](/resources/common#lookupuserbyidsrequest)
- [LookupUserByQueryRequest](/resources/common#lookupuserbyqueryrequest)
- [LookupUserRequest](/resources/admin#lookupuserrequest)
- [LookupUserResponse](/resources/admin#lookupuserresponse)
- [MessageAckRequest](/resources/channels#messageackrequest)
- [MessageAttachmentFlags](/resources/common#messageattachmentflags)
- [MessageAttachmentResponse](/resources/common#messageattachmentresponse)
- [MessageAuthorType](/resources/common#messageauthortype)
- [MessageBaseResponseSchema](/resources/common#messagebaseresponseschema)
- [MessageCallResponse](/resources/common#messagecallresponse)
- [MessageContentType](/resources/common#messagecontenttype)
- [MessageEmbedChildResponse](/resources/common#messageembedchildresponse)
- [MessageEmbedResponse](/resources/common#messageembedresponse)
- [MessageEmbedType](/resources/common#messageembedtype)
- [MessageFlags](/resources/common#messageflags)
- [MessageListResponse](/resources/users#messagelistresponse)
- [MessageReactionResponse](/resources/common#messagereactionresponse)
- [MessageReferenceResponse](/resources/common#messagereferenceresponse)
- [MessageReferenceType](/resources/common#messagereferencetype)
- [MessageReportCategoryEnum](/resources/common#messagereportcategoryenum)
- [MessageResponseSchema](/resources/channels#messageresponseschema)
- [MessageSearchIndexingResponse](/resources/common#messagesearchindexingresponse)
- [MessageSearchResponse](/resources/search#messagesearchresponse)
- [MessageSearchResultsResponse](/resources/common#messagesearchresultsresponse)
- [MessageSearchScope](/resources/common#messagesearchscope)
- [MessageShredRequest](/resources/admin#messageshredrequest)
- [MessageShredResponse](/resources/admin#messageshredresponse)
- [MessageShredStatusNotFoundResponse](/resources/common#messageshredstatusnotfoundresponse)
- [MessageShredStatusProgressResponse](/resources/common#messageshredstatusprogressresponse)
- [MessageShredStatusRequest](/resources/admin#messageshredstatusrequest)
- [MessageShredStatusResponse](/resources/admin#messageshredstatusresponse)
- [MessageSnapshotResponse](/resources/common#messagesnapshotresponse)
- [MessageSortField](/resources/common#messagesortfield)
- [MessageSortOrder](/resources/common#messagesortorder)
- [MessageStickerResponse](/resources/common#messagestickerresponse)
- [MfaBackupCodeResponse](/resources/common#mfabackupcoderesponse)
- [MfaBackupCodesRequest](/resources/users#mfabackupcodesrequest)
- [MfaBackupCodesResponse](/resources/users#mfabackupcodesresponse)
- [MfaSmsRequest](/resources/auth#mfasmsrequest)
- [MfaTicketRequest](/resources/auth#mfaticketrequest)
- [MfaTotpRequest](/resources/auth#mfatotprequest)
- [MutualGuildResponse](/resources/common#mutualguildresponse)
- [MyGuildMemberUpdateRequest](/resources/guilds#myguildmemberupdaterequest)
- [NcmecSubmissionStatusResponse](/resources/admin#ncmecsubmissionstatusresponse)
- [NcmecSubmitResultResponse](/resources/admin#ncmecsubmitresultresponse)
- [NodeStatsResponse](/resources/admin#nodestatsresponse)
- [NSFWLevel](/resources/common#nsfwlevel)
- [OAuth2AuthorizationResponse](/resources/common#oauth2authorizationresponse)
- [OAuth2AuthorizationsListResponse](/resources/oauth2#oauth2authorizationslistresponse)
- [OAuth2ConsentResponse](/resources/oauth2#oauth2consentresponse)
- [OAuth2IntrospectResponse](/resources/oauth2#oauth2introspectresponse)
- [OAuth2MeResponse](/resources/oauth2#oauth2meresponse)
- [OAuth2RedirectURICreateType](/resources/common#oauth2redirecturicreatetype)
- [OAuth2RedirectURIUpdateType](/resources/common#oauth2redirecturiupdatetype)
- [OAuth2TokenResponse](/resources/oauth2#oauth2tokenresponse)
- [OAuth2UserInfoResponse](/resources/oauth2#oauth2userinforesponse)
- [OkResponse](/resources/reports#okresponse)
- [PackCreateRequest](/resources/packs#packcreaterequest)
- [PackDashboardResponse](/resources/packs#packdashboardresponse)
- [PackInviteCreateRequest](/resources/invites#packinvitecreaterequest)
- [PackInviteMetadataResponse](/resources/common#packinvitemetadataresponse)
- [PackInviteResponse](/resources/common#packinviteresponse)
- [PackSummaryResponse](/resources/packs#packsummaryresponse)
- [PackUpdateRequest](/resources/packs#packupdaterequest)
- [PasswordChangeCompleteRequest](/resources/users#passwordchangecompleterequest)
- [PasswordChangeStartResponse](/resources/users#passwordchangestartresponse)
- [PasswordChangeTicketRequest](/resources/users#passwordchangeticketrequest)
- [PasswordChangeVerifyRequest](/resources/users#passwordchangeverifyrequest)
- [PasswordChangeVerifyResponse](/resources/users#passwordchangeverifyresponse)
- [PasswordType](/resources/common#passwordtype)
- [PermissionOverwriteCreateRequest](/resources/channels#permissionoverwritecreaterequest)
- [PhoneAddRequest](/resources/users#phoneaddrequest)
- [PhoneNumberType](/resources/common#phonenumbertype)
- [PhoneSendVerificationRequest](/resources/users#phonesendverificationrequest)
- [PhoneVerifyRequest](/resources/users#phoneverifyrequest)
- [PhoneVerifyResponse](/resources/users#phoneverifyresponse)
- [PreloadMessagesRequest](/resources/users#preloadmessagesrequest)
- [PreloadMessagesResponse](/resources/users#preloadmessagesresponse)
- [PriceIdsResponse](/resources/premium#priceidsresponse)
- [PublicUserFlags](/resources/common#publicuserflags)
- [PurgeGuildAssetErrorSchema](/resources/common#purgeguildasseterrorschema)
- [PurgeGuildAssetResultSchema](/resources/common#purgeguildassetresultschema)
- [PurgeGuildAssetsRequest](/resources/admin#purgeguildassetsrequest)
- [PurgeGuildAssetsResponseSchema](/resources/admin#purgeguildassetsresponseschema)
- [PushSubscribeRequest](/resources/users#pushsubscriberequest)
- [PushSubscribeResponse](/resources/users#pushsubscriberesponse)
- [PushSubscriptionItemResponse](/resources/common#pushsubscriptionitemresponse)
- [PushSubscriptionsListResponse](/resources/users#pushsubscriptionslistresponse)
- [ReactionUsersListResponse](/resources/channels#reactionuserslistresponse)
- [ReadStateAckBulkRequest](/resources/read_states#readstateackbulkrequest)
- [RefreshSearchIndexRequest](/resources/admin#refreshsearchindexrequest)
- [RefreshSearchIndexResponse](/resources/admin#refreshsearchindexresponse)
- [RegisterRequest](/resources/auth#registerrequest)
- [RelationshipNicknameUpdateRequest](/resources/users#relationshipnicknameupdaterequest)
- [RelationshipResponse](/resources/users#relationshipresponse)
- [RelationshipTypePutRequest](/resources/users#relationshiptypeputrequest)
- [RelationshipTypes](/resources/common#relationshiptypes)
- [ReloadAllGuildsResponse](/resources/admin#reloadallguildsresponse)
- [ReloadGuildRequest](/resources/admin#reloadguildrequest)
- [ReloadGuildsRequest](/resources/admin#reloadguildsrequest)
- [RenderSpoilers](/resources/common#renderspoilers)
- [ReorderConnectionsRequest](/resources/common#reorderconnectionsrequest)
- [ReportAdminResponseSchema](/resources/admin#reportadminresponseschema)
- [ReportGuildRequest](/resources/reports#reportguildrequest)
- [ReportMessageRequest](/resources/reports#reportmessagerequest)
- [ReportResponse](/resources/reports#reportresponse)
- [ReportStatus](/resources/common#reportstatus)
- [ReportType](/resources/common#reporttype)
- [ReportUserRequest](/resources/reports#reportuserrequest)
- [ReserveVisionarySlotRequest](/resources/admin#reservevisionaryslotrequest)
- [ResetPasswordRequest](/resources/auth#resetpasswordrequest)
- [ResolveReportRequest](/resources/admin#resolvereportrequest)
- [ResolveReportResponse](/resources/admin#resolvereportresponse)
- [RevokeRequestForm](/resources/oauth2#revokerequestform)
- [RtcRegionResponse](/resources/channels#rtcregionresponse)
- [SavedMessageEntryListResponse](/resources/users#savedmessageentrylistresponse)
- [SavedMessageEntryResponse](/resources/common#savedmessageentryresponse)
- [SaveMessageRequest](/resources/users#savemessagerequest)
- [ScheduleAccountDeletionRequest](/resources/admin#scheduleaccountdeletionrequest)
- [ScheduledMessageAllowedMentionsSchema](/resources/common#scheduledmessageallowedmentionsschema)
- [ScheduledMessageReferenceSchema](/resources/common#scheduledmessagereferenceschema)
- [ScheduledMessageResponseSchema](/resources/users#scheduledmessageresponseschema)
- [SearchAuditLogsRequest](/resources/admin#searchauditlogsrequest)
- [SearchGuildsRequest](/resources/admin#searchguildsrequest)
- [SearchGuildsResponse](/resources/admin#searchguildsresponse)
- [SearchReportsRequest](/resources/admin#searchreportsrequest)
- [SearchReportsResponse](/resources/admin#searchreportsresponse)
- [SearchUsersRequest](/resources/admin#searchusersrequest)
- [SearchUsersResponse](/resources/admin#searchusersresponse)
- [SendPasswordResetRequest](/resources/admin#sendpasswordresetrequest)
- [SentryWebhook](/resources/webhooks#sentrywebhook)
- [SetUserAclsRequest](/resources/admin#setuseraclsrequest)
- [SetUserBotStatusRequest](/resources/admin#setuserbotstatusrequest)
- [SetUserSystemStatusRequest](/resources/admin#setusersystemstatusrequest)
- [SetUserTraitsRequest](/resources/admin#setusertraitsrequest)
- [ShrinkVisionarySlotsRequest](/resources/admin#shrinkvisionaryslotsrequest)
- [ShutdownGuildRequest](/resources/admin#shutdownguildrequest)
- [SlackWebhookRequest](/resources/webhooks#slackwebhookrequest)
- [SnowflakeReservationEntry](/resources/common#snowflakereservationentry)
- [SnowflakeType](/resources/guilds#snowflaketype)
- [SsoCompleteRequest](/resources/auth#ssocompleterequest)
- [SsoCompleteResponse](/resources/auth#ssocompleteresponse)
- [SsoConfigResponse](/resources/common#ssoconfigresponse)
- [SsoStartRequest](/resources/auth#ssostartrequest)
- [SsoStartResponse](/resources/auth#ssostartresponse)
- [SsoStatusResponse](/resources/auth#ssostatusresponse)
- [StickerAnimationOptions](/resources/common#stickeranimationoptions)
- [StreamPreviewUploadBodySchema](/resources/channels#streampreviewuploadbodyschema)
- [StreamUpdateBodySchema](/resources/channels#streamupdatebodyschema)
- [SuccessResponse](/resources/admin#successresponse)
- [SudoMfaMethodsResponse](/resources/users#sudomfamethodsresponse)
- [SudoVerificationSchema](/resources/users#sudoverificationschema)
- [SuspiciousActivityFlags](/resources/common#suspiciousactivityflags)
- [SwapVisionarySlotsRequest](/resources/admin#swapvisionaryslotsrequest)
- [SystemChannelFlags](/resources/common#systemchannelflags)
- [SystemDmJobResponse](/resources/admin#systemdmjobresponse)
- [TempBanUserRequest](/resources/admin#tempbanuserrequest)
- [TenorCategoryTagResponse](/resources/common#tenorcategorytagresponse)
- [TenorFeaturedResponse](/resources/common#tenorfeaturedresponse)
- [TenorGifResponse](/resources/common#tenorgifresponse)
- [TenorRegisterShareRequest](/resources/common#tenorregistersharerequest)
- [TerminateSessionsRequest](/resources/admin#terminatesessionsrequest)
- [TerminateSessionsResponse](/resources/admin#terminatesessionsresponse)
- [ThemeCreateRequest](/resources/themes#themecreaterequest)
- [ThemeCreateResponse](/resources/themes#themecreateresponse)
- [TicketResponse](/resources/reports#ticketresponse)
- [TimeFormatTypes](/resources/common#timeformattypes)
- [TokenRequest](/resources/oauth2#tokenrequest)
- [TransferGuildOwnershipRequest](/resources/admin#transferguildownershiprequest)
- [TriggerGuildArchiveRequest](/resources/admin#triggerguildarchiverequest)
- [TriggerUserArchiveRequest](/resources/admin#triggeruserarchiverequest)
- [UnlinkPhoneRequest](/resources/admin#unlinkphonerequest)
- [UnsignedInt64Type](/resources/common#unsignedint64type)
- [UpdateConnectionRequest](/resources/common#updateconnectionrequest)
- [UpdateFavoriteMemeBodySchema](/resources/saved_media#updatefavoritememebodyschema)
- [UpdateGuildFeaturesRequest](/resources/admin#updateguildfeaturesrequest)
- [UpdateGuildNameRequest](/resources/admin#updateguildnamerequest)
- [UpdateGuildSettingsRequest](/resources/admin#updateguildsettingsrequest)
- [UpdateGuildVanityRequest](/resources/admin#updateguildvanityrequest)
- [UpdateSuspiciousActivityFlagsRequest](/resources/admin#updatesuspiciousactivityflagsrequest)
- [UpdateUserFlagsRequest](/resources/admin#updateuserflagsrequest)
- [UpdateVoiceRegionRequest](/resources/admin#updatevoiceregionrequest)
- [UpdateVoiceRegionResponse](/resources/admin#updatevoiceregionresponse)
- [UpdateVoiceServerRequest](/resources/admin#updatevoiceserverrequest)
- [UpdateVoiceServerResponse](/resources/admin#updatevoiceserverresponse)
- [UrlResponse](/resources/billing#urlresponse)
- [UserAdminResponseSchema](/resources/common#useradminresponseschema)
- [UserAuthenticatorTypes](/resources/common#userauthenticatortypes)
- [UserContactChangeLogEntrySchema](/resources/common#usercontactchangelogentryschema)
- [UserFlags](/resources/common#userflags)
- [UserGuildSettingsResponse](/resources/users#userguildsettingsresponse)
- [UserGuildSettingsUpdateRequest](/resources/users#userguildsettingsupdaterequest)
- [UserMutationResponse](/resources/admin#usermutationresponse)
- [UsernameSuggestionsRequest](/resources/auth#usernamesuggestionsrequest)
- [UsernameSuggestionsResponse](/resources/auth#usernamesuggestionsresponse)
- [UsernameType](/resources/users#usernametype)
- [UserNoteResponse](/resources/users#usernoteresponse)
- [UserNotesRecordResponse](/resources/users#usernotesrecordresponse)
- [UserNoteUpdateRequest](/resources/users#usernoteupdaterequest)
- [UserNotificationSettings](/resources/common#usernotificationsettings)
- [UserPartialResponse](/resources/users#userpartialresponse)
- [UserPremiumTypes](/resources/common#userpremiumtypes)
- [UserPrivateResponse](/resources/users#userprivateresponse)
- [UserProfileFullResponse](/resources/users#userprofilefullresponse)
- [UserReportCategoryEnum](/resources/common#userreportcategoryenum)
- [UserSessionResponse](/resources/common#usersessionresponse)
- [UserSettingsResponse](/resources/users#usersettingsresponse)
- [UserSettingsUpdateRequest](/resources/users#usersettingsupdaterequest)
- [UserStatusType](/resources/common#userstatustype)
- [UserTagCheckResponse](/resources/users#usertagcheckresponse)
- [UserThemeType](/resources/common#userthemetype)
- [UserUpdateWithVerificationRequest](/resources/users#userupdatewithverificationrequest)
- [ValidationErrorCodeSchema](/resources/common#validationerrorcodeschema)
- [ValidationErrorItem](/resources/common#validationerroritem)
- [VerifyAndCreateConnectionRequest](/resources/common#verifyandcreateconnectionrequest)
- [VerifyEmailRequest](/resources/auth#verifyemailrequest)
- [VerifyUserEmailRequest](/resources/admin#verifyuseremailrequest)
- [VisionarySlotOperationResponse](/resources/admin#visionaryslotoperationresponse)
- [VisionarySlotSchema](/resources/common#visionaryslotschema)
- [VoiceRegionAdminResponse](/resources/common#voiceregionadminresponse)
- [VoiceRegionWithServersResponse](/resources/common#voiceregionwithserversresponse)
- [VoiceServerAdminResponse](/resources/common#voiceserveradminresponse)
- [WebAuthnAuthenticateRequest](/resources/auth#webauthnauthenticaterequest)
- [WebAuthnAuthenticationOptionsResponse](/resources/auth#webauthnauthenticationoptionsresponse)
- [WebAuthnChallengeResponse](/resources/users#webauthnchallengeresponse)
- [WebAuthnCredentialListResponse](/resources/users#webauthncredentiallistresponse)
- [WebAuthnCredentialResponse](/resources/common#webauthncredentialresponse)
- [WebAuthnCredentialUpdateRequest](/resources/users#webauthncredentialupdaterequest)
- [WebAuthnMfaRequest](/resources/auth#webauthnmfarequest)
- [WebAuthnRegisterRequest](/resources/users#webauthnregisterrequest)
- [WebhookCreateRequest](/resources/webhooks#webhookcreaterequest)
- [WebhookReceivedResponse](/resources/billing#webhookreceivedresponse)
- [WebhookResponse](/resources/webhooks#webhookresponse)
- [WebhookTokenResponse](/resources/webhooks#webhooktokenresponse)
- [WebhookTokenUpdateRequest](/resources/webhooks#webhooktokenupdaterequest)
- [WebhookType](/resources/common#webhooktype)
- [WebhookUpdateRequest](/resources/webhooks#webhookupdaterequest)
- [WellKnownFluxerResponse](/resources/instance#wellknownfluxerresponse)
- [AdminArchiveResponseSchemaSubjectType](#adminarchiveresponseschemasubjecttype)
- [ListArchivesRequestSubjectType](#listarchivesrequestsubjecttype)
- [PurgeGuildAssetResultSchemaAssetType](#purgeguildassetresultschemaassettype)
- [SearchAuditLogsRequestSortBy](#searchauditlogsrequestsortby)
- [SearchAuditLogsRequestSortOrder](#searchauditlogsrequestsortorder)
- [MessageShredStatusProgressResponseStatus](#messageshredstatusprogressresponsestatus)
- [SearchReportsRequestSortBy](#searchreportsrequestsortby)
- [NcmecSubmissionStatusResponseStatus](#ncmecsubmissionstatusresponsestatus)
- [RefreshSearchIndexRequestIndexType](#refreshsearchindexrequestindextype)
- [SystemDmJobResponseStatus](#systemdmjobresponsestatus)
- [ChannelUpdateTextRequestType](#channelupdatetextrequesttype)
- [ChannelOverwriteRequestType](#channeloverwriterequesttype)
- [ChannelUpdateVoiceRequestType](#channelupdatevoicerequesttype)
- [ChannelUpdateCategoryRequestType](#channelupdatecategoryrequesttype)
- [ChannelUpdateLinkRequestType](#channelupdatelinkrequesttype)
- [ChannelUpdateGroupDmRequestType](#channelupdategroupdmrequesttype)
- [MessageResponseSchemaType](#messageresponseschematype)
- [DonationCheckoutRequestCurrency](#donationcheckoutrequestcurrency)
- [DonationCheckoutRequestInterval](#donationcheckoutrequestinterval)
- [GuildUpdateRequestSplashCardAlignment](#guildupdaterequestsplashcardalignment)
- [GuildUpdateRequestMfaMethod](#guildupdaterequestmfamethod)
- [GuildMemberSearchRequestSortBy](#guildmembersearchrequestsortby)
- [RevokeRequestFormTokenTypeHint](#revokerequestformtokentypehint)
- [PriceIdsResponseCurrency](#priceidsresponsecurrency)

